Author’s response to reviews

Title: Determinants of smoking initiation among women in five European countries: a cross-sectional survey

Authors:

- Debora Lee Oh (deboraoh@usc.edu)
- Julia E Heck (jeheck@ucla.edu)
- Carolyn Dresler (Carolyn.Dresler@arkansas.gov)
- Shane Allwright (sllwright@tdc.ie)
- Margaretha Haglund (margaretha.haglund@fhi.se)
- Carolyn Dresler (Carolyn.Dresler@arkansas.gov)
- Shane Allwright (sllwright@tdc.ie)
- Margaretha Haglund (margaretha.haglund@fhi.se)
- Sara Sanchez Del Mazo (sara.sanchez@ki.se)
- Eva Kralikova (eva.kralikova@lf1.cuni.cz)
- Isabelle Stucker (stucker@vif.inserm.fr)
- Elizabeth Tamang (etamang@ulss15.pd.it)
- Ellen R Gritz (egritz@mdanderson.org)
- Mia Hashibe (mia.hashibe@utah.edu)

Version: 3 Date: 31 January 2010

Author’s response to reviews:

Dear Ms. Abigail Brown,

Thank you again for your consideration of our manuscript. We have added the following statement to the methods section of the manuscript: “This study was approved by the International Agency for Research on Cancer ethical review committee.” We have also made minor changes to the formatting of the document to conform to the journal’s style.

Please let us know if there are any further concerns. We look forward to hearing from you.

Yours sincerely,
Mia Hashibe
Division of Public Health
Department of Family & Preventive Medicine University of Utah School of Medicine
375 Chipeta Way, Suite A
Salt Lake City, UT, 84108
Email: mia.hashibe@utah.edu
Phone: 801-587-3034
Fax: 801-587-3353